Author's response to reviews

Title: Transversal analysis of public policies on user fees exemptions in six West African countries

Authors:

Valéry Ridde (valery.ridde@umontreal.ca)
Ludovic Queuille (queuille@help-ev.de)
Yamba Kafando (odnafak@yahoo.fr)
Emilie Robert (emilie.robert.3@umontreal.ca)

Version: 8 Date: 7 October 2012

Dear Editor

We are pleased to send you a revised version of our article, in which we have tried to address the revisions required by reviewer and you. We hope it now meets the requirements of BMC Health Service Research.

Valéry Ridde